Author's response to reviews

Title: Efficacy of serial C-reactive protein and newborn scale of sepsis for screening of neonatal sepsis at Muhimbili National Hospital, Dar es Salaam

Authors:

Martha F Mkony (mmkony@gmail.com)
Mucho M Mizinduko (mucho_m2002@yahoo.co.uk)
Augustine Massawe (dr.augustine.massawe@gmail.com)
Mecky Matee (mateemecky@yahoo.com)

Version: 2 Date: 12 July 2014

Author's response to reviews:

Response to reviewers comments

1. Grammatical errors corrected
2. unit of measurements on the tables added
3. ROC curves computed for comparison
4. MICs values were not used in this study